Author’s response to reviews

Title: "I beg you…breastfeed the baby, things changed": Infant feeding experiences among Ugandan mothers living with HIV in the context of evolving guidelines to prevent perinatal transmission

Authors:

Emma Dunkley (emma.dunkley@gmail.com)

Scholastic Ashaba (sashaba@must.ac.ug)

Bridget Burns (bfburns@mgh.harvard.edu)

Kasey O'Neil (koneil7@mgh.harvard.edu)

Naomi Sanyu (j_sanomi@yahoo.com)

Cecilia Akatukwasa (cakatukwasa@gmail.com)

Jasmine Kastner (jasminekastner@gmail.com)

Nicole Berry (nsb4@sfu.ca)

Christina Psaros (cpsaros@mgh.harvard.edu)

Lynn Matthews (ltmatthews@mgh.harvard.edu)

Angela Kaida (angela_kaida@sfu.ca)

Version: 1 Date: 04 Dec 2017

Author’s response to reviews:

Dear Editor:

Thank you for reviewing our manuscript “I beg you…breastfeed the baby, things changed”: Infant feeding experiences among Ugandan mothers living with HIV in the context of evolving guidelines to prevent perinatal transmission.”
We have revised the manuscript to reflect the important points raised by the reviewers. We thank the reviewers for their constructive comments and suggestions, and believe that these have strengthened the manuscript.

We have summarized and detailed the revisions in the attached Response to Reviewers. We have also included two versions of the revised manuscript, one with tracked changes and one clean.

We trust that we have addressed all of the concerns and suggestions offered by the reviewers. On behalf of my co-authors, I would like to thank you for your interest in our manuscript. Please do not hesitate to contact me if you have any questions or comments.